Author's response to reviews

Title: Vasorelaxant effect of Prunus yedoensis Matsum bark

Authors:

Kyungjin Lee (dostudy@naver.com)
Inhye Ham (iham@khu.ac.kr)
Gabsik Yang (gabsigi@hanmail.net)
Mihwa Lee (mythio@naver.com)
Youngmin Bu (ymbu@khu.ac.kr)
Hocheol Kim (hckim@khu.ac.kr)
Ho-Young Choi (hychoi@khu.ac.kr)

Version: 4 Date: 18 January 2013

Author's response to reviews: see over
Revise letter

To the Editorial Officer:

*BMC Complementary and Alternative Medicine*

- Thank you for your advices, we changed manuscript as you indicated.

1. Competing interests

- We inserted Competing Interests section in our manuscript. (page 12)

2. Author details

- We also inserted Author details section in our manuscript. (page 13)

Sincerely,

Ho-Young Choi K.M.D., Ph.D., Professor

Department of Herbology,

College of Korean Medicine, Kyung Hee University,

1 Hoegi dong, Dongdaemoon-gu, Seoul 130-701, Republic of Korea

Tel: +82 2 961 9372; Fax: +82 2 965 9372; E-mail address: hychoi@khu.ac.kr